Author's response to reviews

Title: Is Prior Oral Contraceptive Use Associated with Improved Survival in Ovarian Cancer Patients?

Authors:

Aminah Jatoi (jatoi.aminah@mayo.edu)
Nathan Foster (foster.nathan@mayo.edu)
Kimberly Kalli (kalli.kimberly@mayo.edu)
Robert Vierkant (vierkant.robert@mayo.edu)
Zinnia Zhang (zzhang26@illinois.edu)
Melissa Larson (Larson.melissa@mayo.edu)
Brooke Fridley (fridely.brooke@gmail.com)
Ellen Goode (ellen.goode@mayo.edu)

Version: 4 Date: 23 September 2015

Author's response to reviews: see over
September 22, 2015

Prof Sophie F Derchain  
Editor  
BMC Cancer

RE: MS: 7669614161961149  
Research Article: Is Prior Oral Contraceptive Use Associated with Improved Survival in Ovarian Cancer Patients?  
Aminah Jatoi, Nathan Foster, Kimberley Kalli, Robert Vierkant, Zinnia Zhang, Melissa Larson, Brooke Fridley and Ellen Goode  
BMC Cancer (Section: Epidemiology, prevention and public health)

Dear Professor Derchain,

Thank you very much for your completed review of the manuscript referenced above. Thank you also for your comment, “We feel that your manuscript is suitable for publication in the journal.”

In an effort to address the editorial comments that you have raised, we provide a point-by-point response to these other issues, as seen directly below. Changes to the manuscript are highlighted in yellow.

COMMENT: Please provide an author contribution details in the authors’ contribution section.

REPLY: We have now done this.

COMMENT: Please include a competing interest section after discussion.

REPLY: We have now done this. Please note that the authors declare that they have no competing interests.

COMMENT: Please include an acknowledgment section after the authors contribution.

REPLY: We have now provided an acknowledgement in the paper. We have obtained verbal permission from the one person who has been acknowledged.

COMMENT: Please clarify the body responsible for permitting clinical data extraction from the electronic medical records.
REPLY: Karin Goodman, APRN, CNP extracted data from each patient’s clinical record. We have now acknowledged her in the acknowledgement section.

Thank you again for your acceptance of our manuscript. We are honored to have our work published in your journal.

Sincerely,

Aminah Jatoi, M.D.